Author's response to reviews

Title: Barriers to the implementation of preconception care guidelines as perceived by general practitioners: A qualitative study

Authors:

Danielle Mazza (danielle.mazza@monash.edu)
Anna Chapman (anna.chapman@monash.edu)
Susan Michie (s.michie@ucl.ac.uk)

Version: 2 Date: 2 January 2013

Author's response to reviews: see over